Bequest motives in a life-cycle model with intergenerational interactions *

Loretti I. Dobrescu[†], Fedor Iskhakov[‡]

PRELIMINARY AND INCOMPLETE

December 31st, 2013

Abstract

This paper revisits the debate on altruism vs exchange motive for bequest and studies to what extent the saving behavior of elderly Europeans is different in the presence or absence of adult children. To this purpose, we develop a structural model with two overlapping generations, namely elderly parents and their adult children. For each generation, we formulate a separate lifecycle model in which individuals consume and save. Children care about their elderly parents' health and may choose to support them via money transfers and time assistance. Parents, on the other hand, must cover the health costs resulting from heterogeneous health and medical spending shocks, and they can do so via formal insurance (purchased beforehand), informal insurance (provided through time and money transfers by their children) and out-of-pocket. We join the two lifecycle models in a dynamic game between parents and children, which we show has a unique Markov perfect equilibrium. We estimate the model on SHARE data using the simulated method of moments. Preliminary results show a significant bequest motive for savings in Europe. In an altruistic world, children have a considerable incentive to provide both time and financial help, and this greatly impacts the parents' savings behavior. Moreover, health, medical spending and health insurance also appear to be crucial in determining the old age saving patterns. Finally, counterfactual experiments show that considering only the strategic motive for bequest cannot explain the slow wealth decumulation in old age or the children's transfers patterns.

Key Words: bequest, health, health insurance, intergenerational transfers, family interactions, dynamic game, method of moments.

JEL Classification: D1, D31, E27, H31, H51, I1.

^{*}The authors are especially grateful to Orazio Attanasio, Chris Carroll, Mariacristina De Nardi, Eric French, Michalis Halissos, John Jones, Alberto Motta, Makoto Nakajima, Jose-Victor Rios-Rull, John Rust, Xiaodong Fan, and participants at the 2013 Australasian Meeting of the Econometric Society, 2013 Annual NBER Summer Institute Workshop on Aggregate Implications of Microeoconomic Behavior for helpful comments and suggestions. Research support from the Australian Research Council Centre of Excellence in Population Ageing Research (CEPAR), under project CE110001029 is gratefully acknowl-edged.

[†]CEPAR and School of Economics, UNSW, Sydney 2052, Australia. dobrescu@unsw.edu.au [‡]CEPAR, UNSW, Sydney 2052, Australia. f.iskhakov@unsw.edu.au

1 Introduction

As the world grows old, the question of why do the elderly dissave so slowly is becoming increasingly important. Do they keep wealth to support themselves in retirement or to leave bequests to their children? The economic literature has investigated both these precautionary savings and bequest explanations extensively (Kotlikoff and Summers, 1981; Bernheim et al., 1985; Dynan et al., 2002). Among the sources of risk that induce the elderly to engage in precautionary savings, health and medical spending have long been recognized as two of the most significant (Hubbard et al., 1994, 1995; Palumbo, 1999; Dynan et al., 2004; De Nardi et al., 2010). Recently, the availability of health insurance, both formal and informal, also proved to have a sizeable effect on wealth decumulation in old age (Guariglia and Rossi, 2004; Dobrescu, 2012).

In terms of bequests, the key issue of whether they result from a deliberate motive or from unpredictable death has not yet garnered an univoqual answer (see Hurd, 1987, 1989; Hendricks, 2002; Kopczuk and Lupton, 2007; Laitner and Sonnega, 2012; Lokwood, 2012).¹ And if people do save to leave bequests, the question is why. They could be motivated by pure altruism, e.g., parents care about the well-being of their grown up children (Becker, 1974; Barro, 1974; Laitner, 1992; 2001). Or there could be a strategic (or exchange) motive where parent-to-child emotional and social ties favor non-market exchanges that may generate bequests, e.g., bequests may emerge as payments to heirs for personal services rendered (Bernheim et al., 1985; Kotlikoff and Spivak, 1981). Finally, another alternative is the "warm glow" motive, where parents derive pleasure from making transfers to their adult children, but that pleasure is not specifically dependent upon the children's utility gain (Blinder, 1974).

The current paper revisits the 'altruism vs. exchange motive for bequest' and studies to what extent the saving behavior of elderly Europeans (and relatedly their bequest motive) is different in the presence or absence of adult children.² To this purpose, we develop a structural model with two overlapping generations, namely elderly parents and

¹For instance, Hurd (1987) finds that people with children decummulate their wealth faster than people without children, while Hurd (1989) finds the bequest motive to be economically trivial. On the contrary, Kopczuk and Lupton (2007) find that 79 percent of households with children have a bequest motive compared with 63 percent of childless households.

 $^{^{2}}$ We note that individuals with adult children, unlike those with underage children, do not have generally higher expenses than the childless, but they do have a potential bequest motive in their saving behavior.

their adult children. In our life cycle model, children care about the health of their parents (Johnson and Lo Sasso, 2000; Bonsang, 2007). As a result, they consume and save, but may also choose to support their parents via money transfers or time spent providing care to them. Parents, on the other hand, consume, save and also have to cover the health costs resulting from heterogeneous health and medical spending shocks. There are three options to fund health costs: i) formal insurance that has to be purchased beforehand, ii) informal insurance provided through time and money transfers by their children, and iii) out-of-pocket. We also include a "warm-glow"-type of parental altruism to study the extent to which such a bequest motive can explain the slow dissaving observed in old age. Finally, we join the two models in a dynamic game between parents and children.

This is obviously not the first attempt to construct an overlapping generations model with bequests and intergenerational transfers (Laitner, 1992, 2001; De Nardi, 2004; Fuster et al., 2002; Nishiyama, 2002). Up to our best knowledge, however, this is the first study to develop a dynamic game between parents and children, in which the interaction involves intergenerational transfers of both time and money and the size of bequeathable wealth. Thus, this paper also contributes to the extensive literature on *inter vivos* transfers and on their impact on savings (Kotlikoff, 1988; Gale and Scholz, 1994; Rendall and Bahchieva, 1998) and bequests (Brown, 2006; 2007).³

We estimate the model on data from the Survey of Health, Ageing and Retirement in Europe (SHARE), using the simulated method of moments (SMM) for three European regions: Scandinavia - Denmark and Sweden, Central Europe - Austria, Belgium, France, Germany, Netherlands, Switzerland, and the Mediterranean - Italy, Spain and Greece.⁴ This country classification corresponds to several interesting patterns observed in the SHARE data. Specifically, there seems to be a very clear North-South gradient in formal and informal care provision: formal insurance appears more prevalent in Denmark and Sweden, whereas the Mediterraneans rely more on informal arrangements. When looking more closely at informal care, time transfers appear to be much more common than financial transfers, with higher levels of time assistance in Italy, Spain and Greece.⁵

³Rendall and Bahchieva (1998) for instance argue that without transfers from children (and relatives or friends), the U.S. elderly poverty rate would be double. In return, parents are more likely to transfer their bequest to children who provide them with regular care (Brown, 2006).

⁴See Gullestad and Segalen (1997) for institutional differences between the three regions.

⁵In SHARE, 32.8 percent of the single individuals aged 65 and above receive help in the form of time,

Preliminary results from our structural model show a significant bequest motive for savings in Europe. In an altruistic world, children have a considerable incentive to provide both time and financial help, and this greatly impacts the parents' savings behavior. Moreover, health, medical spending and health insurance also appear to be crucial in determining the old age saving patterns. Finally, counterfactual experiments show that considering only the strategic motive for bequest cannot explain the slow wealth decumulation in old age or the children's transfers patterns.

Final results (and more findings) to be provided once the simulations have converged.

From a policy perspective, the reason why the elderly dissave so slowly and how they use their wealth will determine not only their own well-being, but will also affect the living standards of their children, the resources available and the level of investment capital in the economy. In order to predict the impact of policy changes on future wealth accumulation, it is important to understand not only the disposition of wealth, but also why and how people save (or dissave) in the first place. In this sense, a model that is capable of explaining the choices of European elderly can significantly improve our understanding and design of reforming policies.

The remainder of the paper is organized as follows: Section 2 gives an overview of the health care systems and their funding in Europe. Section 3 develops the dynamic model, and Section 4 describes the data. Section 5 presents the estimation method, using the SMM methods. Results are illustrated in Section 6, and experiments are conducted in Section 7. Section 8 concludes.

2 Some European facts

Health care arrangements vary enormously across Europe, providing an ideal setting to study the effect of institutional and cultural differences across countries. Table 1 provides selected statistics related to the prevalence of formal versus informal, and public versus private health care funding across the 11 European countries in SHARE.

A quick glance reveals that the public policy in these countries is based on the principle of health care funded by the state or by social insurance, made available to all individuals which is almost seven times higher than the prevalence of financial help. and covering most of the major health shock. There are three types of public healthcare systems, following a north-south gradient: i) national health services in Scandinavia (Denmark and Sweden), ii) social-insurance systems in Central Europe (Austria, Belgium, France, Germany, the Netherlands and Switzerland), and iii) a mixed type of systems that can be seen as 'third way' (Freeman, 2000), established in the early 1980s in the Mediterranean (Italy, Spain and Greece).

The common feature of these systems is that they provide almost universal health care, across two dimensions. On the one hand, being financed through taxation or contributions from employers and employees, participation in the public system is usually mandatory.⁶ On the other hand, these systems cover all severe (life-threatening) medical conditions, offering comprehensive benefits that account for more than 70 percent of the total health care expenditure.⁷

The existence of near universal public coverage in Europe reduces the basic need for additional insurance. However, the exclusion of certain health services from the statutory coverage, like specialist or diagnostic outpatient services, drugs, dental care, medical appliances, glasses, alternative medicine, occasional choice of better or faster inpatient care for important interventions (Paccagnella et al., 2013) has led to the development of a private health insurance market. As a result, 32.7% of the roughly 3,800 elderly single individuals represented in SHARE Wave 1 hold private health insurance (as shown in Table 2). This contract can be offered as a short-term or as a long-term arrangement, with premiums almost always set as a flat rate per month (or year) and used to finance health care costs. The norm for private health insurance in the European Union is short-term contracts (typically annual), with roughly \notin 700 premiums on average in our sample. The lowest prevalence of private health insurance is registered in the Mediterranean countries and Sweden where less than 6.5% of respondents hold health insurance. In Sweden however the small size of the private insurance market is traditionally attributed to the generosity of the public system: health care is predominantly financed through national

 $^{^{6}}$ Around 99 percent of the European population is covered by these schemes. The exceptions are Germany and the Netherlands, where people with income above a certain threshold have to be privately insured.

⁷The exceptions are Switzerland that has a private health care system, compulsory for everyone, and Greece that has a mix of national health, social and private insurance system.

and local general taxation⁸ that cover roughly 82% of the total health expenditure.

Despite the relatively high prevalence of formal health insurance, the benefits paid accounted for less than 5% of the total expenditure on health in most countries (Dobrescu, 2012). Thus, private expenditure is largely generated by out-of-pocket payments (as seen in Table 2), with above 70% of the elderly in our sample having annual median out-of-pocket expenses of roughly $\in 600$.

In recent years, several empirical studies have documented the strong correlation between the amount of care provided via the formal schemes and the care supplied informally by close relatives or neighbors. For instance, Bolin et al. (2008) finds that informal and formal home care are substitutes, while informal care is a complement to doctor and hospital visits. These relationships also differ according to a European north-south gradient: compared to those residing in Italy, Spain or Greece, the negative effect of informal care on formal home care is significantly lower for Central Europeans and absent for Danish and Swedish. On the other hand, Bonsang (2009) shows that informal care is an effective substitute for long-term care only as long as the needs of the elderly are low and require unskilled type of care.

Informal care can be provided in the form of time and financial transfers given to parents. Elderly people who receive financial help from their children represent a small minority of less than 4% in all countries (see Table 3), except in Greece (15.1%), Austria (5.1%) and Spain (4.9%). Variation across countries is relatively large in terms of financial transfers prevalence, but regime patterns are difficult to discern. The story is quite different however in terms of the amounts of financial help received. For instance, Danish and Swedish are clearly the ones that offer the highest amount of financial support (both in median or average terms). On the other hand, in the Southern group – with the exception of Italy – the amount of financial transfers from children to parents is not significantly different from the Continental average level of transfers: €950 in Greece and €1,000 in Spain vs. roughly €950 in Continental Europe.

Regarding time assistance, on average, the elderly singles receive above 50% of their informal care from relatives and friends (as shown in Table 4). Children provide the remaining 50% of care, except in Greece, where they provide almost 73%. The country

⁸In Sweden local taxes in 2003-2004 were approximately 72% (Glenngård et al. 2005).

patterns of social support follow only partially the proposed distinction between the three types of welfare regimes. For the prevalence of support, these regimes seem not to matter: on average, 24.1% of the elderly singles in our sample receive help from children, and this proportion is significantly higher in Denmark, Sweden, Belgium, Germany and Greece than in Switzerland, Italy and Spain. On the other hand, the intensity of social support matches well the three regime clusters. The lowest group in terms of the average number of help hours is that of Denmark, Sweden and the Netherlands, while the highest one is Italy, Spain and Greece, with all the continental countries in between.

All these country-specific factors have a strong impact on the elderly wealth decumulation patterns. Thus, the model will be estimated separately for the Mediterranean, Central Europe and Scandinavia.

3 The Model

We model the two generations of parents (p) and children (k) as unitary decision making agents who live separately. When making their every period choices of consumption, purchase of formal health insurance or coverning health expences out-of-pocket, parents eventually spend down their wealth, thus decreasing the size of potential bequest for the children. In realization of this fact, as well as for the altruistic reasons, children may decide to provide financial support for their parents or to provide care for them by sacrificing some of their working hours.

3.1 Lifecycle model for the parents

When modeling the parent's generation we focus on consumption, health insurance and savings decisions, and disregard choices related to labour supply, timing of retirement and household dynamics. Thus, we focus on single retired individuals, and fix the retirement age at 65, with certain death occuring at the age of 100, i.e. $t \in [65, 100]$.

Preferences: In each time period the parent seeks to maximize the expected utility over the remainder of her life, by choosing the levels of consumption C_t and formal health insurance premium f_t , while children are providing a time transfer T_t and a money transfer M_t to help with illness episodes and other expenses. The future is discounted with a constant discount factor $\beta^p \in [0, 1]$, and within period utility is given by

$$u_{t}^{p}(m_{t}, C_{t}^{p}, T_{t}) = \delta_{t} \frac{\left[(C_{t}^{p})^{\theta_{t}} + \alpha_{t} \cdot (T_{t})^{\theta_{t}} \right]^{\frac{1 - \gamma^{p}}{\theta_{t}}} - 1}{1 - \gamma^{p}},$$
(1)

where $1/\gamma^p$ is the intertemporal elasticity of substitution, $1/(1 - \theta_t)$ is the elasticity of substitution between consumption C_t^p and the value of time assistance provided by children T_t^{9} , and α_t is scale parameter. We note that the time dependent coefficients θ_t and α_t vary as health deteriorates over time. This implies that in worse health states children's assistance may become relatively more important than physical consumption.

When the parent dies, all bequeathable wealth (net of inheritance taxes) B_t is transferred to her children. The utility from such accidental bequest $u_b(B_t)$ is given by

$$u_b(B_t) = b \frac{(B_t + k)^{1 - \gamma^p} - 1}{1 - \gamma^p},$$
(2)

where b is the intensity of the bequest motive and k determines the extent to which bequests are luxury goods (De Nardi et al., 2010).

In each period, the parent can be in one of four possible health states (Dobrescu, 2012): i) good health, $m_t = \mathbb{G}$; ii) fair health (some medical problems, but no need for long-term care), $m_t = \mathbb{F}$; iii) poor health (invalidity or some form of long-term care required), $m_t = \mathbb{I}$; iv) death, $m_t = \mathbb{D}$. We treat symbols \mathbb{G} , \mathbb{F} , \mathbb{I} and \mathbb{D} as scalar parameters and impose the following normalization and ordering constraint:

$$0 = \mathbb{D} < \mathbb{I} < \mathbb{F} < \mathbb{G} = 1.$$
(3)

We assume that risk aversion γ^p is time invariant and that the coefficient δ_t is given by (Palumbo, 1999)

$$\delta_t = \begin{cases} 1 + m_t & \text{if } 0 < m_t \le 1, \\ 0 & \text{if } m_t = 0, \end{cases}$$
(4)

i.e., higher utility is associated with better health, and utility drops to zero at death.

⁹We impose a strict restriction $0 < \theta_t < 1$ to ensure that first, even with no assistance time $T_t = 0$, utility of consumption is well defined, and second, with any amount of time assistance, utility of zero consumption is infinitely negative (i.e., marginal utility of consumption approaches plus infinity when $T_t = 0$ and $C_t^p \to 0^+$).

The parameter α_t denotes the value of time assistance relative to consumption, and takes the form

$$\alpha_t = \alpha \cdot (2 - m_t), \quad \alpha \ge 0, \tag{5}$$

allowing for consumption to become relatively less important than children's assistance in bad health states.

Finally, we assume that elasticity of substitution between consumption and the value of children's time assistance falls with health, namely

$$\theta_t = \theta_0 + \theta \cdot m_t, \quad \theta_0, \theta \ge 0. \tag{6}$$

In other words, more consumption has to be given up to retain the same level of utility in worse health if children choose not to increase the time support for parents because of, for example, less costly options (like hiring a caretaker) have to be chosen.

Health transitions: There are two sources of uncertainty in the parent's problem. First, health status m_t evolves as a Markov chain with transition probability

$$p_{jk}^{m}(t, s_{t}^{p}) = \Pr(m_{t+1} = k | m_{t} = j, t, s_{t}^{p}),$$
(7)

that allows the next period health to depend on age t, current health m_t and parent's end-of-period wealth s_t^p . We do not model investments in health directly, but instead we use end-of-period wealth to proxy for them. As it will be clear below, this also makes the parent's health transitions dependent on monetary transfers from children, and thus children may affect parent's health dynamics. To keep the model computationally tractable, we disregard other factors that may influence health transitions.¹⁰

We parametrize the transition probability matrix $[p_{jk}^m(t, s_t^p)]_{j,k \in \{\mathbb{G},\mathbb{F},\mathbb{I},\mathbb{D}\}}$ by assuming that age and wealth effects on health can be expressed with two separate adjustment

¹⁰One other important factor that might affect health transitions is education. In Europe, however, Avendano et al. (2009) find however that education is not significantly associated with transitions to or out of (long-term) illness. As for disability, switching from primary school to postgraduate levels matters only in Central Europe and the Mediterranean, and only for transitions into this state (not out of it). The relative association between education and health also diminishes with age (Huisman et al., 2005a) and even disappears after retirement (due to stable incomes and universal health insurance coverage). Since we focus on retirees, not accounting for education at this stage seemed reasonable.

matrices, $A_1(t)$ and $A_2(s_t^p)$, as follows

$$\left[p_{jk}^{m}(t,s_{t}^{p})\right]_{j,k\in\{\mathbb{G},\mathbb{F},\mathbb{I},\mathbb{D}\}} = A_{0} + A_{1}(t) + A_{2}(s_{t}^{p}),$$
(8)

where A_0 is the initial health transition probability matrix at age 64. We impose the following parametrization to the initial transition probability matrix A_0 :

$$A_{0} = \begin{bmatrix} c_{1}(1-\delta_{0}) & \frac{1-c_{1}}{2}(1-\delta_{0}) & \frac{1-c_{1}}{2}(1-\delta_{0}) & \delta_{0} \\ \frac{1-c_{2}}{4}(1-\delta_{0}) & c_{2}(1-\delta_{0}) & \frac{3(1-c_{2})}{4}(1-\delta_{0}) & \delta_{0} \\ \frac{1-c_{3}}{4}(1-\delta_{0}) & \frac{3(1-c_{3})}{4}(1-\delta_{0}) & c_{3}(1-\delta_{0}) & \delta_{0} \\ 0 & 0 & 0 & 1 \end{bmatrix},$$
(9)

where δ_0 is calibrated to the survival probability between age 64 and 65.11

Following Ameriks et al. (2005), we model the health dependence on age in (8) with an age adjustment matrix

$$A_{1}(t) = \begin{bmatrix} -c_{4}\tau^{\alpha_{m}} \frac{c_{4}c_{5}c_{6}\tau^{\alpha_{m}}}{1+c_{5}+c_{5}c_{6}} \frac{c_{4}c_{5}\tau^{\alpha_{m}}}{1+c_{5}+c_{5}c_{6}} \frac{c_{4}\tau^{\alpha_{m}}}{1+c_{5}+c_{5}c_{6}} \\ 0 & -c_{4}\tau^{\alpha_{m}} \frac{c_{4}c_{5}\tau^{\alpha_{m}}}{1+c_{5}} \frac{c_{4}\tau^{\alpha_{m}}}{1+c_{5}} \\ 0 & 0 & -c_{4}\tau^{\alpha_{m}} c_{4}\tau^{\alpha_{m}} \\ 0 & 0 & 0 & 0 \end{bmatrix},$$
(10)

which shifts the transition probability mass towards worse health, relative to the transitions at age 65. The variable $\tau = t - 65$ denotes the number of years since retirement, while α_m is a fixed parameter that allows for faster than linear shifting in health as one becomes older. The other three parameters in (10) have the following roles: c_4 controls the transition from invalidity to death as age increases, c_5 determines how much more likely is death relative to invalidity when in fair or good health, and c_6 determines an individual's chance to persist in good health.

¹¹When choosing the specification of the initial health transition probability matrix A_0 we were constrained *first* by the limitations of our data that precluded the estimation of health transitions between 64 and 65 directly, and *second* by considerations of computational tractability of our model and identification of the large number of parameters in the health transition process.

Finally, the wealth adjustment matrix $A_2(s_t^p)$ is given by

$$A_{2}(s_{t}^{p}) = \begin{bmatrix} c_{7}\ln(s_{t}^{p}+1) & -c_{7}\ln(s_{t}^{p}+1) & 0 & 0\\ c_{8}\ln(s_{t}^{p}+1) & -c_{8}\ln(s_{t}^{p}+1) & 0 & 0\\ 0 & c_{9}\ln(s_{t}^{p}+1) & -c_{9}\ln(s_{t}^{p}+1) & 0\\ 0 & 0 & 0 & 0 \end{bmatrix},$$
(11)

where the level of parents' end-of-period wealth shifts the probability mass in health transitions towards transitions to better health.¹² We assume that higher wealth increases the chances for healthy to stay in good health (first row) and decreases the chance of the next worse health stata (\mathbb{F}). For the individuals in fair (\mathbb{F}) or poor (\mathbb{I}) health, higher wealth provides better opportunity for recovery (i.e., higher chances of transitioning into \mathbb{G} or \mathbb{F} , respectively). The last row in (11) contains no adjustments because \mathbb{D} is an absorbing state.

Health spending: The second source of uncertainty in the parents problem besides health transitions is out-of-pocket medical spending $oop_t(m_t)$. We define oop_t as the part of total health costs $h(m_t)$ not covered by formal insurance $F_t(f_{t-1}, m_t)$. Out-of-pocket costs may only appear in bad health states $(m_t < 1 = \mathbb{G})$, except that in the event of death $(m_t = 0 = \mathbb{D})$ only a one time deterministic funeral cost h_0 is incurred, i.e.,

$$oop_t(m_t) = \begin{cases} h_0 & \text{if } m_t = 0 = \mathbb{D}, \\ \max\{0, \ h(m_t) - F_t(f_{t-1}, m_t) + \psi_t\} & \text{if } m_t \in (0, 1), \\ 0 & \text{if } m_t = 1 = \mathbb{G}. \end{cases}$$
(12)

Health cost $hc(m_t)$ is an exogenous deterministic process and health spending shock ψ_t follows a Gaussian autoregressive process

$$\ln \psi_t = (1 - \rho_\psi) \ln \overline{\psi} + \rho_\psi \ln \psi_{t-1} + \varepsilon_t^p, \ \varepsilon_t^p \backsim N(0, \sigma_{\varepsilon^p}^2),$$
(13)

where ρ_{ψ} , $\overline{\psi}$ and σ_{ε^p} are fixed parameters. Without loss of generality, we assume that $\psi_t = 0$ when $m_t \in \{0, 1\}$.

 $^{^{12}}$ Despite being theoretically possible, the adjusted values for health transition probabilities never reach 0 or 1 during the simulations or estimation of the model.

Budget constraint and health insurance: The model allows for two links between each pair of consequent years. The first link is due to the standard consumption-savings process: parents' end-of-period wealth s_t^p at time t grows with constant risk-free interest rate r, so that at period t + 1, they start with wealth level $s_t^p(1 + r)$.

The second link is due to formal medical insurance. Following Dobrescu (2012), we represent a typical contract in the following simplified way. In exchange for paying an annual premium f_t , individuals have a share of their next period medical costs covered by formal health insurance $F_{t+1}(f_t, m_{t+1})$. Besides the private component, this contract includes a minimum health care consumption floor $\overline{F}(m_{t+1})$ provided by the public health system, as follows:

$$F_{t+1}(f_t, m_{t+1}) = \begin{cases} \overline{F}(m_{t+1}) + \frac{\omega}{m_{t+1}} f_t, \text{ if } m_{t+1} \in (0, 1), \\ 0, & \text{if } m_{t+1} \in \{0, 1\}, \end{cases}$$
(14)

where $\frac{\omega}{m_{t+1}}$ is the inverse of the loading factor.¹³

Any medical expenditures in excess of the formal health coverage must be paid outof-pocket. To help with expenses children can provide to their elderly parents not only time assistance T_t , but also financial help M_t . Let w_t^p denote the wealth of parents *net of* the transfer from children. Assuming that resources can only be spent on consumption and formal insurance premium, the end-of-period wealth is

$$s_t^p = w_t^p + M_t - C_t^p - f_t, (15)$$

and the intertemporal budget constraint is given by

$$w_{t+1}^{p} = \max \left\{ c_{\text{floor}}^{p}, \ (w_{t}^{p} + M_{t} - C_{t}^{p} - f_{t})(1+r) + y_{t+1} - oop_{t+1}(m_{t+1}) \right\}$$

$$= \max \left\{ c_{\text{floor}}^{p}, s_{t}^{p} \cdot (1+r) + y_{t+1} - oop_{t+1}(m_{t+1}) \right\},$$
(16)

where y_t is pension benefit in period t and $oop_t(m_t)$ is out-of-pocket medical spending defined in (12). Wealth must satisfy the borrowing constraint $s_{t+1}^p \ge 0$, which eliminates the possibility of parents dying in debt. Because out-of-pocket spending is not bounded

¹³The loading factor captures the health-specific ratio of formal premium to coverage, net of any administrative costs.

from above, we include a possibility for personal bankruptcy by introducing a consumption floor denoted c_{floor}^p at the level that corresponds to $\frac{1}{1000}$ of the average pension level. Because according to (16) the pension payment is spent for covering out-of-pocket health spending, personal bunkruptsy is effectively punished by at least one period of consumption at level c_{floor}^p .

Recursive form of the parent's problem: The Bellman equation of the parent's problem takes the form (for t < 100):

$$V_{t}^{p}(w_{t}^{p}, m_{t}, \psi_{t}; M_{t}, T_{t}) = \max_{C_{t}^{p}, f_{t}} \{ u_{t}^{p}(m_{t}, C_{t}^{p}, T_{t}) + \beta^{p}(1 - p_{m_{t}\mathbb{D}}^{m}) E_{m,\varepsilon^{p}} \left[V_{t+1}^{p}(w_{t+1}^{p}, m_{t+1}, \psi_{t+1}; M_{t+1}, T_{t+1}) | m_{t+1} \neq \mathbb{D} \right] + \beta^{p} p_{m_{t}\mathbb{D}}^{m} u_{b}(B_{t+1}) \}$$

$$(17)$$

subject to $C_t^p \ge 0$, $f_t \ge 0$ and the credit constraint $w_t^p + M_t - C_t^p - f_t \ge 0$. The expectation is taken over health transitions and health spending shock ε_t^p in (13). Note that the last two arguments of value function $V_t^p(w_t^p, m_t, \psi_t; M_t, T_t)$ are marked out to emphasize the fact that they are determined by the actions of the children. When t = 100 the value function is given by

$$V_t^p(w_t^p, m_t, \psi_t; M_t, T_t) = \max_{C_t^p} \left\{ u_t^p(m_t, C_t^p, T_t) + \beta^p u_b(B_t) \right\},$$
(18)

subject to $C_t^p \ge 0$ and the credit constraint $w_t^p - C_t^p \ge 0$. Because the decision maker dies with certainty after the termination period, it is never optimal to "waste" resources by purchasing formal insurance, i.e., $f_{100} = 0$.

3.2 Lifecycle model for the children

We model children as one representative unit with unitary preferences, i.e. as if each parent had either none or a single child. We focus on modelling children's choices of hours of labour supply and transfers to parents after their parents retire at age 65. When parents die, children inherit the remained of parent's wealth, and do not have to make any transfers thereafter. Children's interactions with parent in the model are driven by both strategic and altruistic motives. By providing time and money transfers children can affect the rate of parent's wealth decumulation in the anticipation of larger bequest. The altruistic motive is related to the fact that children care about their parents' health, and this plays a crucial role in the decision to help with money and/or time (Johnson and Lo Sasso, 2000; Bonsang, 2007).

Preferences: In each period t, children's within period utility is given by

$$u_t^k(C_t^k, m_t) = \frac{\left(C_t^k\right)^{1-\gamma^k} - 1}{1-\gamma^k} + \eta m_t,$$
(19)

where $1/\gamma^k$ is the intertemporal elasticity of substitution of children's consumption C_t^k , while η quantifies the altruism motive in children's preferences.

Transfers to parents: Besides consumption, in every period t, children choose how much time to spend helping their parents and how much money to transfer to them (i.e., they choose M_t and T_t). These transfers will be primarily directed to cover parents' health spending, but we do not condition them on parent's health status m_t .

Monetary transfers M_t are made out of children's beginning-of-period wealth w_t^k , thus

$$w_t^k = C_t^k + M_t + s_t^k, (20)$$

where, analogous to parents problem, s_t^k denotes end-of-period wealth.

When it comes to time transfers T_t , we focus on the trade-off between wage earnings and potential bequest, and therefore assume that the time children choose to spend assisting their parents is taken out of their working time (Bolin et al., 2008; Johnson and Lo Sasso, 2000; Carmichael and Charles, 1998, 2003a,b; Heitmueller, 2007; Heitmueller and Inglis, 2007). For simplicity, we disregard children's leisure.¹⁴ Let L denote per period working time endowment, and H_t the time actually spent at work. Then the time budget is

$$H_t + T_t = L. (21)$$

¹⁴SHARE does not collect data on time use, and so we do not have any information on children's time allocation between work, leisure and help provided to their elderly parents. The literature on informal care however has extensively documented the negative impact of informal care on carers work hours. Thus, assuming that children's time assistance would reduce their work time is both in line with previous empirical evidence (see Bolin et al. 2008 for evidence on Europe) and it ultimately avoids the identification issues raised by the lack of data.

Wage equation: We use a traditional Mincer (1958) type specification for the children's wage equation. Specifically, the wage W_t follows an age profile with idiosyncratic log-normal shocks, namely

$$\ln W_t = \lambda_0 + \lambda_1 (t - \Delta) + \lambda_2 (t - \Delta)^2 + \lambda_3 e du + \varepsilon_t^k, \qquad (22)$$

where Δ is the offset of the child's age from the parent's age, *edu* denotes the time invariant education level of children, and $\varepsilon_t^k \sim N(0, \sigma_{\varepsilon^k}^2)$.

Budget constraint and bequest: Similarly to parents, children earn the risk-free interest rate r on their savings, while their incomes consist of wage earnings and anticipated bequest. We make a common assumption that wage is paid after a waiting time of one period.¹⁵ Therefore, next period wealth for children is given by

$$w_{t+1}^{k} = (w_{t}^{k} - C_{t}^{k} - M_{t})(1+r) + W_{t+1}H_{t} + B_{t+1}(m_{t+1})$$

$$= s_{t}^{k} \cdot (1+r) + W_{t+1}H_{t} + B_{t+1}(m_{t+1}),$$
(23)

where $B_{t+1}(m_{t+1})$ is the bequest the children expect to receive if the parent dies. The size of the bequest is

$$B_{t+1} = \begin{cases} 0 & \text{if } m_{t+1} > 0, \\ (1-\varsigma) \max\left\{0, s_t^p (1+r) + y_{t+1} - h_0\right\} & \text{if } m_{t+1} = 0 = \mathbb{D}, \end{cases}$$
(24)

where $m_{t+1} > 0$ is realized with probability $1 - p_{m_{t-1}\mathbb{D}}^m$ and $m_t = 0$ is realized with probability $p_{m_{t-1}\mathbb{D}}^m$. The parameter ς denotes the effective inheritance tax rate on wealth transferred as bequest.

Recursive form of the children's problem: The Bellman equation of the children's problem takes the form (for t < 100)

$$V_t^k(w_t^k, edu; m_t, s_t^p) = \max_{C_t^k, M_t, T_t} \left\{ u_t^k(C_t^k, m_t) + \beta^k E_{\varepsilon^k} \left[V_{t+1}^k(w_{t+1}^k, edu; m_{t+1}, s_{t+1}^p) \right] \right\}, \quad (25)$$

¹⁵The essential part of this assumption is that labour supply choices are made before the wage shock is realized. This allows for treatment of wage shock as idiosyncratic and simplifies the numerical solution because shocks do not have to be included in the state space. An alternative interpretation is that wages are paid in the end of the period with a multiplicator $\frac{1}{1+r}$ and idiosyncratic shocks are relabelled with one period offset.

subject to $C_t^k \ge 0$, credit constraint $s_t^k = w_t^k - C_t^k - M_t \ge 0$ and time budget $0 \le T_t \le L$. Note that similarly to parent's problem, the last two arguments of the value function $V_t^k(w_t^k, edu; m_t, s_t^p)$ are marked out to emphasize that they are determined by the actions of the parents.

The future in children's problem is discounted with discount factor β^k that may not equal parent's discount factor β^p . For simplicity, we disregard the possibility of parents outliving their children and so, survival probability is not present in children's Bellman equation.

Denote by $t_d \leq 100 - \Delta$ the age of the children when parents die, (i.e. the first time when $m_t = \mathbb{D}$ is realized). In this period, bequests are realized according to (24), and from this period on the action space of the children is reduced to a consumptionsavings choice only: for $t \geq t_d$, children solve a 'cake eating' problem with the size of the cake increasing every period by a random amount corresponding to full wage (22). We use Carroll's (2006) endogeneous grid method to compute the numerical solution for this reduced problem for children, and use the computed value functions as the children's terminal values when the parents die. Effectively, we can terminate the children's problem at time t_d and start backwards induction for the children's problem from the terminal value at that time.

3.3 Dynamic intergenerational game

While making their individual choices, parents and children account for the impact of these choices on the other party. Thus, in our model parents and children find themselves in a dynamic intergenerational finite horizon game.

Common state space and common knowledge: As it follows from the Bellman equations of both parents and children's problems in (17) and (25), the actions of the opponent enter the state space of each problem. When we consider the two problems together within the intergenerational game, these actions are endogenized, and thus, the state space of the game is given by

$$\Omega_t = \{ w_t^p, m_t, \psi_t, w_t^k, edu, \Delta \}.$$
(26)

We note that the age difference Δ between parents and children enters necessarily into the state space of the game to facilitate the alignment of their life cycles in the unified model, as in (22). Further, we assume that idiosyncratic shocks ε_t^p and ε_t^k to health expenditures and to wages are independent and are common knowledge within the family once they are realized. In other words, the information sets of parents and children are identical at each moment in time, forming a complete information game.

Timing of the events and the order of moves in the game: The sequence of events in the intergenerational game is the following: Period t starts with the realization of (i) the health shock that determines m_t , and (ii) the shock ε_t^k in the wage equation (22). Current incomes (i.e., the pension benefit y_t for parents and the wage earnings W_tH_{t-1} for children) are then immediately realized.

If the parent dies $(m_t = \mathbb{D})$ bequest is realized according to (24), and children's help is no longer needed $(M_t = 0 \text{ and } T_t = 0)$. Children's wealth w_t^k is then fully determined by (23), and they are assigned with a termination value. The game ends.

In case parents survive $(m_t > 0)$, medical spending shock ψ_t is realized, and outof-pocket expenses $oop_t(m_t)$ are determined according to (12), taking into account the formal health insurance f_{t-1} purchased in the previous period. Also, by this time all components in (23), in particular $B_t = 0$, are known and children simultaneously choose time assistance T_t (which determines work hours H_t), consumption C_t^k and money transfers M_t to parents.

We assume that children's choices become known to the parents *before* they have to move, and thus parents know not only their wealth w_t^p given by (16), but also the amount M_t transferred from children, which together define their consumable resources. Parents then choose consumption C_t^p and purchase formal insurance at premium f_t . Period t ends after consumption has taken place, and wealth decreases to its end-of-period values s_t^p and s_t^k . The game continues to the next period.

Equilibrium concept: The fact that in each period parents observe children's decisions $(C_t^k, M_t \text{ and } T_t)$ before making their own choices $(C_t^p \text{ and } f_t)$, together with the assumption of complete information, ensures that the information sets of the players are singletons. Thus, the intergenerational game belongs to the class of finite dynamic games of perfect information. Consequently we only consider the pure strategy equilibria of the game, and following the Markov nature of the dynamic problems at hand, we adopt the concept of Markov Perfect Equilibrium (MPE).

Definition 1 The pure strategies MPE equilibrium in the dynamic intergenerational game is given by a pair of strategies $(\varphi_t^p, \varphi_t^k)$

$$\begin{aligned} \varphi_t^p : \Omega_t \to (C_t^p, f_t), \\ \varphi_t^k : \Omega_t \to (C_t^k, M_t, T_t), \end{aligned}$$
(27)

which map the state space into the action space of parents and children for each t and constitute the solutions of the Bellman equations of these two problems given in (18-17) and (25), conditional on the endogenously determined actions of the opponent.

3.4 Solution method

According to Definition 1, the solution of our model is a pair of policy functions that jointly solve the system of Bellman equations given in (17-18) and (25). To find the solution, we develop a backwards induction algorithm that sequentially searches for Markov perfect equilibria in subgames spanning over parents' ages t to 100, where we allow t to iteratively decrease from 100 to 65. Consequently, we first solve for an equilibrium in a "terminal game" when t = 100, which forms the base of the induction. Then given the value functions $V_{t+1}^p(\Omega_{t+1})$, $V_{t+1}^k(\Omega_{t+1})$ and the policy functions φ_{t+1}^p , φ_{t+1}^k we solve for all equilibria in period t subgame by computing the mutual best responses in period t conditional on strategies φ_{t+1}^p , φ_{t+1}^k to be used at period t + 1 and beyond.¹⁶ Proposition 1 below ensures the uniqueness of the fixed point in the mutual best responses of children and parents arising from the maximization problems in (18-17) and (25), thus establishing the uniqueness of the equilibrium in every subgame $t \in \{100, 99, ..., 65\}$, and eventually the uniqueness of the MPE in the whole game.

Characterization of stage equilibrium: The equilibrium in a given period t is comprised of a pair of policy functions (27) that defines the correspondence between the points

¹⁶Theorem 5 in Iskhakov, Rust and Schjerning (2013) provides justification of this algorithm in computing subgame equilibria.

in the state space (26) and the decision vectors $(C_t^p, f_t, C_t^k, M_t, T_t)$. We construct this correspondence by solving for the optimal decisions (fixed point in mutual best responses) in every point of the *discretized* state space.

We discretize the continuous dimensions in the state space Ω_t using both fixed and endogenous grids, following the endogenous grid point method developed by Carroll (2006). Specifically, the grid over ψ_t is fixed, and the grids over wealth variables w_t^p and w_t^k are determined in the solution algorithm as explained below. The state variables m_t , eduand Δ are discrete.

Given a point in the discretized state space, the optimal decisions of parents and children are characterized by a complex system of non-linear equations (28).¹⁷ This system consists of the Euler equations for parents' and children's problems (18-17) and (25) with respect to each of the five continuous decision variables.¹⁸ In the numerous cases

 17 The full system of first order conditions derived in sections A.1 and A.2 in the Appendix is

$$\begin{split} \frac{\partial u_{t}^{p}(m_{t},C_{t}^{p},T_{t})}{\partial C_{t}^{p}} &= \beta^{p}(1+r)\tilde{E}_{m,\varepsilon} \left[\frac{\partial u_{t+1}^{p}(m_{t+1},C_{t+1}^{p},T_{t+1})}{\partial C_{t+1}^{p}} \middle| m_{t+1} \neq \mathbb{D} \right] \\ &+ \beta^{p}(1+r)p_{m_{t}\mathbb{D}}^{m}(1-\varsigma)u_{b}^{\prime}(B_{t+1}) \\ &+ \beta^{p}\sum_{m^{\prime}\neq\mathbb{D}} \frac{\partial p_{m_{t}m^{\prime}}^{m}}{\partial s_{t}^{p}} E_{\varepsilon}V_{t+1}^{p}(w_{t+1}^{p},m^{\prime},\psi_{t+1};M_{t+1},T_{t+1}), \qquad (a) \\ 0 &= -(1+r)\tilde{E}_{m,\varepsilon} \left[\frac{\partial u_{t+1}^{p}(m_{t+1},C_{t+1}^{p},T_{t+1})}{\partial C_{t+1}^{p}} \middle| m_{t+1} \neq \mathbb{D} \right] - (1+r)p_{m_{t}\mathbb{D}}^{m}(1-\varsigma)u_{b}^{\prime}(B_{t+1}) \\ &- \beta^{p}\sum_{m^{\prime}\neq\mathbb{D}} \frac{\partial p_{m_{t}m^{\prime}}^{m}}{\partial s_{t}^{p}} E_{\varepsilon}V_{t+1}^{p}(w_{t+1},m^{\prime},\psi_{t+1};M_{t+1},T_{t+1}) \\ &+ \frac{w}{1}\tilde{E}_{\varepsilon} \left[\frac{\partial u_{t+1}^{p}(\varepsilon,C_{t+1}^{p},T_{t+1})}{\partial C_{t+1}^{p}} \middle| m_{t+1} = \mathbb{I} \text{ and } \exp(\varepsilon_{t+1}^{p}) > \Lambda(\mathbb{I},f_{t}) \right] \\ &+ \frac{w}{\mathbb{P}}\tilde{E}_{\varepsilon} \left[\frac{\partial u_{t+1}^{k}(\varepsilon,C_{t+1}^{k},T_{t+1})}{\partial C_{t+1}^{p}} \middle| m_{t+1} = \mathbb{F} \text{ and } \exp(\varepsilon_{t+1}^{p}) > \Lambda(\mathbb{F},f_{t}) \right] \qquad (b) \\ \frac{\partial u_{t}^{k}(C_{t}^{k},m_{t})}{\partial C_{t}^{k}} = \beta^{k}(1+r)E_{m,\varepsilon} \left[\frac{\partial u_{t+1}^{k}(C_{t+1}^{k},m_{t+1})}{\partial C_{t+1}^{k}} \right] - (1+r)\frac{\partial s_{t}^{p}}{\partial M_{t}}\tilde{E}_{\varepsilon} \left[\frac{\partial u_{t+1}^{k}(C_{t+1}^{k},m_{t+1})}{\partial C_{t+1}^{k}} \middle| m_{t+1} = \mathbb{D} \right] \\ &- \frac{\partial s_{t}^{2}}}{\partial M_{t}} \sum_{m'} \frac{\partial p_{m_{t}m'}^{m}}}{\partial s_{t}^{p}}} E_{\varepsilon}V_{t+1}^{k}(w_{t+1}^{k},edu;m',s_{t+1}^{p}), \qquad (d) \end{aligned}$$

where the expectations are taken over idiosyncratic shocks in wage equation ε_t^k , shocks ε_t^p in the health spending equation where indicated by ε and health transitions where indicated by m, $\Lambda(m_{t+1}, f_t)$ is given in (34), and we use the shortcut notation $\check{E}(\circ|\star) = E(\circ|\star)P(\star)$. In addition, 47 corner solutions are characterized by the same system after replacing equations (a-e) in (28) according to the rules given by all combinations of the cases in the following three groups

$$\begin{array}{ll} (a,b)\mapsto (36), w_t^p = C_t^p + f_t, & (e)\mapsto T_t = 0, & (c,d)\mapsto (41), w_t^k = C_t^k + M_t, \\ (a,b)\mapsto (a), f_t = 0, & (e)\mapsto T_t = L, & (c,d)\mapsto (c), M_t = 0, \\ (a,b)\mapsto f_t = 0, C_t^p = w_t^p, & (e) \text{ as is, } & (c,d)\mapsto M_t = 0, C_t^k = w_t^k, \\ (a,b) \text{ as is, } & (c,d) \text{ as is. } \end{array}$$

 $^{18}\mathrm{We}$ do not discretize any of the decision variables in our model except time transfers $T_t.$

of corner solutions, arising from non-negativity and credit constraints in both problems, as well as time budget constraint in children's problem, the system of first order conditions also includes the corresponding binding constraints.

Proposition 1 The MPE equilibrium in the intergenerational bequest game is unique.

Proof. We provide an intuition instead of a rigorous proof here. First we show that the the system (28) has a unique solution in every t given the continuation strategies φ_{t+1}^p , φ_{t+1}^k . Then, because the equilibrium path necessarily passes through the fixed points in mutual best responses in every period, it follows that the whole dynamic MPE equilibrium is unique as well.

To show the former, note that system (28) can be rearranged to yield optimal values of each of the decision variables one by one similar to a triangular system (see Appendix A for the details). Therefore it suffices to show the uniqueness of solution of each equation in (28).

The complete description of the solution algorithm is given in Appendix A.

4 Data

The data we use to estimate the model are drawn from the first two waves of the Survey of Health, Ageing and Retirement in Europe (SHARE).¹⁹ SHARE is a longitudinal, multidisciplinary and cross-national survey representing the population of individuals aged 50 and over in Europe. The first wave of SHARE took place in 2004, with the second one following in 2006. There are 11 countries participating in both waves, namely Austria, Belgium, Denmark, France, Germany, Greece, Italy, the Netherlands, Spain, Sweden and Switzerland. The total number of individuals interviewed in 2004 (Release 3) is 28,517.

Our analysis focuses on the sample of single, non-institutionalized individuals,²⁰ with ages 65 and above. The raw data on this sample cover about 4,500 individuals. When the elderly receive help from co-residing children or other co-residing adults such as siblings, we do not observe the amount of hours they provide. Therefore, we exclude

¹⁹http://www.share-project.org/

 $^{^{20}}$ van Houtven and Norton (2004) point out that the types of care needed by institutionalized individuals differ substantially from those of the elderly living at home. Nursing-home residents for instance may have less access to care from children, and so, we exclude them from the sample. We note however that the institutionalized subsample represents only 2.6% of the full sample we consider.

about 13.7% of the individuals that have adult children living in the same household. As a last restriction, we impose that each parent must have non-negative net worth (Brown, 2005), which leads to a final sample of 3,796 individuals, of whom 921 are men and 2,875 are female. Of these individuals, 3,628 are still alive in 2006.

On the parents side, the four variables of interest for our model are: i) total net worth in PPP adjusted \in (wealth henceforth), ii) total expenditures on non-durables (consumption henceforth), iii) annual voluntary (supplementary) private insurance premium (premium henceforth), and iv) total amount spent out-of-pocket on medical goods and services. Wealth represents the value of all financial and real assets, net of any debts and liabilities (i.e., loan repayment, mortgage, taxes). Pension benefits, on the other hand, include any public and private pensions or invalidity benefits, such as old age, early or pre-retirement pension, disability benefits, survivor and war pension, private annuities, etc. Out-of-pocket medical spending was computed as the total amount spent on drugs, inpatient and outpatient care, as well as on nursing home care, day-care and home care. We note that they include medical expenses incurred during respondent's last year of life.

To insure cross-country comparability, we follow Browning et al. (2003) and calculate total consumption as the sum of the amount spent on food (at home and outside home) and phone bills, adjusted by country-specific weights. These weights represent OLS coefficients of a regression of total non-durable consumption on a subset of household expenses (i.e., groceries, eating out, phone bills and other utilities, transportation, clothing, entertainment, etc.) using national expenditure survey data. We use the IS-TAT2004 survey for Italy, Spain and Greece, and the Dutch Budget Survey (DBS) for the remaining countries.

Finally, the question on the amount spent on premium was only asked in the first SHARE wave. Moreover, the corresponding data clearly reflects the limited ability of older parents to report this cost. Although continuous information on premiums is so-licited, the responses contain many missing values. Given the stability of insurance profiles in old age (Paccagnella et al., 2013), we impute the missing values in Wave 1 and the corresponding values for Waves 2 using the coefficients of age, education, health, wealth and consumption from an OLS regression on existing premium data.²¹

 $^{^{21}}$ The imputation affected 64.7% of the whole sample of individuals reporting wealth and consumption

Given its focus on ageing, SHARE provides only limited information on the respondents' adult children. For instance, there is no indication of children's earnings or consumption value, but there is detailed demographics data on age, gender, marital status, number of children, education and occupation for up to four children. To retrieve the wage and consumption profiles, we use two sets of data, as follows: i) the ISTAT2004 and DBS data, to compute consumption, and ii) the European Union - Statistics on Income and Living Conditions (EU-SILC) data, to compute wages. The procedure employed to obtain the consumption profiles of adult children (of age 25-64) is simple. For each possible combination of the demographic characteristics that are reported in SHARE, we compute the median value of consumption using the ISTAT2004 (for Italy, Spain and Greece) and the DBS data (for the other countries). We then assign to each individual in SHARE the correspondent imputed consumption value obtained from these alternative datasets, based on his/her specific demographic characteristics.²²

To impute the wages of children, we follow a similar procedure to the one used to retrieve consumption for their elderly parents. Specifically, we model the wage profiles as a function of age, gender, marital status, number of children, education, occupation and country of residence. Next, we estimate the wage equation on country-specific data from the EU-SILC and use the coefficients to retrieve the wage profiles of SHARE individuals, based on the common demographic variables available for children.

Importantly, we model children as one representative unit, i.e., as if each parent had either none or a single child. As a result, for parents with more children, we consider as representative the child who provides the most time and/or money transfers, or the oldest if none of the children provide help. If the most time and money transfers come however from different children, we take their median.

One of the advantages of using SHARE is that it contains detailed information on financial transfers and social support from the perspective of the 'parents' aged 65 and above. In this paper we use the information on: (i) the frequency and amount of financial

in Wave 1. Alternative specifications were explored, but the one implemented fitted best the existing data. More details on the imputation procedure are available from the authors.

 $^{^{22}}$ Using this procedure, we manage to match 2,282 individuals that represent 78.7% of the sample of children. We note that this is a conservative matching procedure (i.e., we match across six demographic dimensions), but given the importance of the data profiles for children, we opt for matching accuracy over sample size.

or material gifts or support (other than for shared housing and food) of at least $\in 250$ (or the equivalent in local currency) from children living outside the household; (ii) the frequency and amount (in hours) of social support received in any of the three forms – personal care, practical household help and help with paperwork – from children living outside the household. Both these questions refer to the 12 months prior to the interview.

We note that when asked about the financial transfers' amount, some respondents refuse to answer or encounter difficulties in remembering exact amounts. In these cases, a series of brackets unfolds, with the respondent being asked whether the amount is similar to, lower or higher than or in-between these bracket values. This information can be used to reduce the number of missing values (Christelis, 2011). Thus, we use the SHARE generated data module that contains the imputed values for the three financial transfers reported by each respondent. Finally, among the transfers received from children we select the one provided by the representative child.

In terms of time transfers, for each person from whom the respondent received social support information is collected on: i) the identity of the caregiver and his/her relationship with the respondent, and ii) the frequency of this help and its average amount in hours. Respondents can receive help "almost daily", "almost every week", "almost every month" and "less often". Depending on the answer to this question, respondents are asked about the number of hours on a "typical day/week/month" or "in the last twelve months". In order to obtain the number of hours of support in the last 12 months, we therefore multiply the average number of hours provided daily, weekly and monthly by 365 days, 52 weeks and 12 months, respectively. Receiving time assistance from more than one child is not common: among the households which receive informal home care, only 1.5% report having two or three children providing help. Similar to financial transfers, we construct the variable denoting time assistance from children as equal to the annual hours provided by the representative child.

In addition to constructing moment conditions, we also use the SHARE data to construct the region-specific initial distributions of wealth, age difference (between parents and children), pensions and out-of-pocket medical expenses, by (children's) education and (parents) health to start off our simulations. As a result, each simulated individual will be endowed with a state vector drawn from the joint distribution of the state variables observed in 2004.

5 Calibrations and estimation methodology

The life cycle literature based on European data is guite limited and the institutional differences are potentially significant. Therefore, we estimate most of the model parameters for each region separately, and calibrate only those parameters that appear as instruments for the dynamic programming model (Gourinchas and Parker, 2002; Cagetti, 2003; French and Jones, 2011). We set the real risk-free asset return at $(1 + r) = 1.04^{23}$, α_m at 1.5 (Ameriks et al., 2005) and ς at the levels reported by AGN International (2013). For each European region, we follow Dobrescu (2012) and set the public insurance coverage \overline{F} to the adjusted mean 2004 level of: i) public curative and rehabilitation (CR henceforth) expenditure per capita, if in fair health; ii) public long-term care (LTC henceforth) expenditure per capita, if in poor health. Similarly, the exogenous total medical spending hmatches the adjusted country-specific 2004 level of: i) total CR costs per capita, if in fair health; ii) total LTC per capita, if in poor health; iii) funeral costs, if dead (Bjornerud et al., 2005).²⁴ Since \overline{F} and h represent absolute per capita amounts (OECD, 2006), the adjustment involved re-weighting their value with the Eurostat (2012) share of population reporting good or fair health (for CR costs) and bad or very bad health (for LTC costs). The parameter $\overline{\psi}$ is set to the OECD 2004 PPP adjusted total expenditure on health per capita, while ρ_{ψ} , $\sigma_{\varepsilon^p}^2$, γ^p and β^p are taken from Dobrescu (2012). Finally, δ_0 is calibrated to the average survival probability between age 64 and 65 in Europe using Eurostat and Human Mortality Database ($\delta_0 = 0.00914$ for Mediterranean countries, $\delta_0 = 0.01036$ in Central Europe, and $\delta_0 = 0.01098$ in Scandinavia), whereas the age difference Δ between parents and children is taken from the SHARE data.

The aim of the analysis is to explain the wealth decumulation profiles, as well as the formal and informal insurance decisions in old age. Hence, we match total wealth, consumption, formal premium, out-of-pocket medical spending, amounts and prevalence of time and financial transfers, as well as children's consumption, conditional on age and

 $^{^{23}}$ The average long-term interest rate in 2004 for the Mediterranean countries was 4.2 percent, for Central Europe 3.9 percent and for Scandinavia 4.4 percent (OECD Statistics 2010, Key Economic Indicators).

 $^{^{24}\}mathrm{Further}$ details on health and funeral costs are provided in Appendix B.

education. To calculate the empirical moments, we first break the data into five cohorts and three levels of education.²⁵ The 1st cohort consists of individuals with ages 65-69 in 2004, the 2nd cohort contains ages 70-74, the 3rd cohort contains ages 75-79, the 4th cohort contains ages 80-84, and the 5th cohort, for sample size reasons, contains ages 85+. We use the data for two different years corresponding to the two SHARE waves, namely 2004 and 2006. To calculate the moments of our variables of interest, we take cell medians by cohort and education, for surviving individuals in each calendar year.²⁶

To calculate the simulated moments, for each region, we proceed as follows: To compute the optimal choices, we use the Gauss-Hermite quadrature method to discretize the state space of the shocks. We solve the model numerically by backward induction and simulate wealth, consumption, formal premium, health and medical spending histories for N = 10,000 artificial individuals, using Monte Carlo draws for the two stochastic variables. Additionally, we also simulate consumption, wages and time and money transfers profiles of their children. For each artificial profile, we compute cell medians by cohort and education level conditional on the initial values of the state variables Ω_0 and on model parameters $\phi' = (\theta_0, \theta, \alpha, \omega, b, k, c_{i=\overline{1,9}}, \gamma^k, \beta^k, \eta) \in \mathbb{R}^{18}$.

Finally, parameters are chosen to minimize the difference between these artificial moments and their empirical counterparts. The goodness of fit between the two series is assessed by a χ^2 -test statistic or corresponding p - value. This statistic assesses whether or not the true data moments (m_T) are equal to the realized data moments $(m_n(\Omega_0, \phi'), n = \overline{1, N})$, given the stochastic processes for which the true time series is only one realization. Analytically, as $NT \to \infty$, keeping the number of random sequences fixed, if the weighting matrix W is chosen optimally, then

$$T \cdot \arg\min_{\widetilde{\phi}'} \left[\left(m_T - \frac{1}{N} \sum_{n=1}^N m_n(\mathbf{\Omega}_0, \widetilde{\phi}') \right)' \widehat{W} \left(m_T - \frac{1}{N} \sum_{n=1}^N m_n(\mathbf{\Omega}_0, \widetilde{\phi}') \right) \right] \to \chi^2(j-k),$$

where j is the number of moments, k is the number of estimated parameters and $\phi' \in R^k$ is the unknown parameter vector.²⁷

²⁵We define education via a variable denoting whether respondents have secondary education, finished high school or whether they pursued further education after high school.

 $^{^{26}}$ Cells with less than 10 observations are excluded from the moment conditions. Due to the low prevalence of financial transfers, the medians for this variable were calculated only by education.

²⁷The standard errors of the parameters are obtained using the Newey and West (1994) weighting

6 Estimation results

Our estimation procedure is still currently running, but preliminary results are available from the authors upon request.

7 Counterfactual simulations

To be completed.

8 Conclusions

To be completed.

References

- [1] AGN International. 2013. Gift and Inheritance Tax 2004-2009: A European Comparison, available online at http://www.agn-europe.org/tax/index.html#intl.
- [2] Ameriks, J., Caplin, A., Laufer, S., Van Nieuwerburgh, S. 2011. "The Joy of Giving or Assisted Living? Using Strategic Surveys to Separate Public Care Aversion from Bequest Motives" The journal of finance Vol.LXVI, No.2
- [3] Avendano, M., Jurges, H., Mackenbach, J.P. 2009. "Educational Level and Changes in Health Across Europe: Longitudinal Results From SHARE." Journal of European Social Policy 19(4): 301-16.
- [4] Barro, R.J. 1974. "Are government bonds net worth?" Journal of Political Economy 82: 1095-117.
- [5] Becker, G.S. 1974. "A theory of social interactions." Journal of Political Economy 82: 1063-93.
- [6] Bernhein, B.D., Shleifer, A., Summers, L.H. 1985. "The Strategic Bequest Motive." Journal of Political Economy 93: 1045-76.
- [7] Bjornerud, S., Oliveira Martins, J. 2005. "Disentangling Demographic and Nondemographic Drivers of Health Spending: A Possible Methodology and Data Requirements." OECD Economics Department, Presented at the Joint EC/OECD Workshop 2005.
- [8] Blinder, A.B. 1974. "Toward an Economic Theory of Income Distribution." Cambridge, MA: MIT Press.
- [9] Bolin, K., Lindgren, B., Lundborg, P. 2008. "Informal and Formal Care Among Single-Living Elderly In Europe." Health Economics 17(3): 393-409.
- [10] Bonsang, E. 2007. "How Do Middle-aged Children Allocate Time and Money Transfers to their Older Parents in Europe?" Empirica 34(2): 171-188.
- [11] Bonsang, E. 2009. "Does Informal Care from Children to Their Elderly Parents Substitute for Formal Care in Europe?" Journal of Health Economics 28: 143-54.
- [12] Brown, M. 2006. "Informal care and the division of end-of-life transfers." Journal of Human Resources 41(1): 191-219.

matrix W and the first order derivatives of the moments conditions with respect to each parameter: $se = \frac{1}{\sqrt{T}} (DW^{-1}D')^{-1} \text{ with } D' = plim \left\{ \left. \frac{\partial \left[m_T - \frac{1}{N}m_N(\Omega_0, \phi') \right]}{\partial \phi'} \right|_{\phi' = \tilde{\phi'}} \right\}.$

- [13] Brown, M. 2007. "End-of-life transfers and the decision to care for a parent." Working Paper, Federal Reserve Bank of New York.
- [14] Browning, M., Crossley, T.F., Weber, G. 2003. "Asking Consumption Questions in General Purpose Surveys." Economic Journal 113: F540-F567.
- [15] Cagetti, M. 2003. "Wealth Accumulation Over the Life Cycle and Precautionary Savings." Journal of Business and Economic Statistics 21(3): 339-53.
- [16] Carmichael, F., Charles, S., 1998. "The labour market costs of community care." Journal of Health Economics 17: 747765.
- [17] Carmichael, F., Charles, S., 2003a. "The opportunity costs of informal care: does gender matter?" Journal of Health Economics 22: 781803.
- [18] Carmichael, F., Charles, S., 2003b. "Benefit payments, informal care and female labour supply." Applied Economics Letters 10: 411415.
- [19] Carroll, C.D. 2006. "The method of endogenous gridpoints for solving dynamic stochastic optimization problems." Economic Letters 91: 312-320
- [20] Christelis, D. 2011. Imputation of Missing Data in Waves 1 and 2 of SHARE." SHARE Working Paper 01/2011.
- [21] De Nardi, M., French, E., Jones, J.B. 2010. "Why do the Elderly Save? The Role of Medical Expenses." Journal of Political Economy 118(1): 39-75.
- [22] De Nardi, M. 2004. "Wealth Inequality and Intergenerational Links." Review of Economic Studies 71: 743-68.
- [23] Dobrescu, L.I. 2012. "To Love or to Pay: Savings and Health Care in Older Age." Working Paper, University of New South Wales.
- [24] Dynan, K., Skinner, J., Zeldes, S. P. 2002. "The Importance of Bequests and Life-Cycle Saving in Capital Accumulation: A New Answer." AER Papers and Proceedings 2: 274-8.
- [25] Dynan, K. E., Skinner, J., Zeldes, S. P. 2004. "Do the Rich Save More?" Journal of Political Economy 112(2): 397-444.
- [26] Freeman R. 2000. "The Politics of Health in Europe." Manchester University Press.
- [27] French, E., Jones, J.B. 2011. "The Effects of Health Insurance and Self-insurance on Retirement Behavior." Econometrica 79: 693-732.
- [28] Fuster L., Imrohoroglu A., Imrohoroglu S. 2003. "A Welfare Analysis of Social Security in a Dynastic Framework." International Economic Review 44(4):1247-74
- [29] Gale, W.G., Scholz, J.K. 1994. "Intergenerational Transfers and the Accumulation of Wealth." Journal of Economic Perspectives 8(4): 145-160.
- [30] Glenngaard, A.H., Hjalte, F., Svensson, M., Anell, A., Bankauskaite, V. 2005. "Health systems in transition: Sweden." in Health Systems in Transition, 7(4):1–128.
- [31] Gourinchas, P.O., Parker, J.A. 2002. "Consumption Over the Life Cycle." Econometrica 70(1): 47-89.
- [32] Guariglia, A., Rossi, M. 2004. "Private Medical Insurance and Saving: Evidence from the British Household Panel Survey." Journal of Health Economics 23(4):761-83.
- [33] Gullestad M., Segalen, M. 1997. "Family and Kinship in Europe." London, Washington, Pinter.
- [34] Van Houtven, C.H., Norton, E.C. 2004. "Informal care and health care use of older adults." Journal of Health Economics 23(6): 1159-1180.
- [35] Heitmueller, A. 2007. "The chicken or the egg? Endogeneity in labour market participation of informal carers in England." Journal of Health Economics 26(3): 536-559.
- [36] Heitmueller, A., Inglis, K. 2007. "The earnings of informal carers: wage differentials and opportunity costs." Journal of Health Economics 26: 821841.
- [37] Hendricks, L. 2002. "Intended and Accidental Bequests in a Life-cycle Economy." mimeo.
- [38] Hubbard, R.G., Skinner, J., Zeldes, S.P. 1994. "The Importance of Precautionary Motives in Explaining Individual and Aggregate Saving." Carnegie-Rochester Conference Series on

Public Policy 40: 59-125.

- [39] Hubbard, R.G., Skinner, J., Zeldes, S.P. 1995. "Precautionary Saving and Social Insurance." Journal of Political Economy 103(2).
- [40] Huisman, M., Kunst, A.E., Bopp, M., Borgan, J.K., Borrell, C., Costa, G., Deboosere, P., Gadeyne, S., Glickman, M., Marinacci, C., Minder, C., Regidor, E., Valkonen, T., Mackenbach, J.P. 2005a. "Educational Inequalities in Cause-specific Mortality in Middleaged and Older Men and Women in Eight Western European Populations." Lancet, 365: 493-500.
- [41] Hurd, M.D. 1987. "Savings of the Elderly and Desired Bequests." American Economic Review 77(3): 298-312.
- [42] Hurd, M.D. 1989. "Mortality Risk and Bequests." Econometrica 57(4): 779-813.
- [43] Johnson, R.W., Lo Sasso, A.T. 2000. "The Trade-Off between Hours of Paid Employment and Time Assistance to Elderly Parents at Mid-Life." Washington, DC: The Urban Institute.
- [44] Kopczuk, W., Lupton, J. P. 2007. "To Leave or Not to Leave: The Distribution of Bequest Motives." Review of Economic Studies 74(1): 207-235.
- [45] Kotlikoff, L. J., Summers, L. H. 1981. "The Role of Intergenerational Transfers in Aggregate Capital Accumulation." Journal of Political Economy 59(4): 706-32.
- [46] Kotlikoff, L. 1988. "Intergenerational Transfers and Saving." Journal of Economic Perspectives 2: 41-58.
- [47] Kotlikoff, L.J., Spivak, A. 1981. "The family as an incomplete annuities market." Journal of Political Economy 89: 372-91.
- [48] Laitner, J.P., Sonnega, A. 2012. "Motives for Bequests within the Middle Class." Working Paper, University of Michigan.
- [49] Laitner, J. 1992. "Random earnings differences, lifetime liquidity constraints and altruistic intergenerational transfers." Journal of Economic Theory 58: 135-70.
- [50] Laitner, J. 2001. "Secular changes in wealth inequality and inheritance." Economic Journal 111: 691-721.
- [51] Lockwood, L.M. 2012. "Incidental Bequests: Bequest Motives and the Choice to Self-Insure Late-Life Risks." Working paper, Northwestern University.
- [52] Mincer, J. 1958. "Investment in Human Capita and Personal Income Distribution." Journal of Political Economy 66: 281-302.
- [53] Newey, W.K., West, K.D. 1994. "Automatic Lag Selection in Covariance Matrix Estimation." Review of Economic Studies 61(4): 631-53.
- [54] Nishiyama S. 2002. "Bequests, inter vivos transfers, and wealth distribution." Review of Economic Dynamics 5(4): 892-931.
- [55] Paccagnella, O., Rebba, V., Weber, G. (2013). "Voluntary Private Health Care Insurance Among the Over Fifties: A Comparative Analysis of SHARE Data." Health Economics 22(3): 289315.
- [56] Palumbo, M.G. 1999. "Uncertain Medical Expenses and Precautionary Saving Near the End of the Life Cycle." Review of Economic Studies 66: 395-421.
- [57] Rendall, M., Bahchieva, R. 1998. "An Old-age Motive for Fertility in the United States?" Population and Development Review 24(2): 293-307.
- [58] Westerhout, E., Pellikaan, F. 2005. "Can We Afford to Live Longer in Better Health?", CPB Bureau for Economic Policy Analysis Netherlands Document No. 85.
- [59] Eurostat. The inhabitants of a given area on 1 January of the year 2004. Available at http://epp.eurostat.ec.europa.eu/tgm/table.do?tab=table&init=1&plugin=0&language=en& pcode=tps00001 (data downloaded on [March 7, 2013]).
- [60] Eurostat. 2012. "Self-Perceived Health by Sex, Age and Activity status." http://epp.eurostat.ec.europa.eu/portal/page/portal/statistics/search database
- [61] Human Mortality Database. University of California, Berkeley (USA), and Max Planck

Institute for Demographic Research (Germany). Available at www.mortality.org or www.humanmortality.de (data downloaded on [March 7, 2013]).

- [62] OECD, 2006. "Demography and Health Health Spending and Resources." in OECD in Figures 2006-2007, OECD 2006 - ISBN 9264022635, http://dx.doi.org/10.1787/427077022807.
- [63] OECD, 2006. Projecting OECD Health and Long-Term Expenditures: What Are the Main Drivers?, OECD Economics Department Working Paper 477.
- [64] OECD, 2008, OECD Health Data 2008, http://www.oecd.org/document/27/0,3343,en26493463 140902299111100.html.
- [65] OECD Statistics, 2010. Key Economic Indicators. OECD.Stat extracts. Organization for Economic Cooperation and Development 2010, http://stats.oecd.org.

A Derivations and proofs

A.1 Euler equations in the parent's problem

The parents problem is given in the Bellman equations (17-18). Consider first the case of *interior solution* where the constraint $w_t^p - C_t^p - f_t \ge 0$ does not bind.²⁸ The standard combination of the FOCs for the optimal level of consumption with the envelope condition leads to a standard identity between marginal utility evaluated at the optimal consumption level and the partial derivative of the value function with respect to consumable wealth in the same period. Euler equation that characterizes the optimal levels of consumption in two adjacent time periods, however, is complicated by the utility of bequest and the fact that health transition probabilities are dependent on the consumption level through end-of-period wealth. For t < 100 we have

$$\frac{\partial u_t^p(m_t, C_t^p, T_t)}{\partial C_t^p} = \beta^p (1+r) \check{E}_{m,\varepsilon} \left[\frac{\partial u_{t+1}^p(m_{t+1}, C_{t+1}^p, T_{t+1})}{\partial C_{t+1}^p} \middle| m_{t+1} \neq \mathbb{D} \right]
+ \beta^p (1+r) p_{m_t \mathbb{D}}^m (1-\varsigma) u_b'(B_{t+1})
+ \beta^p \sum_{m' \neq \mathbb{D}} \frac{\partial p_{m_t m'}^m}{\partial s_t^p} E_{\varepsilon} V_{t+1}^p(w_{t+1}^p, m', \psi_{t+1}; M_{t+1}, T_{t+1}),$$
(30)

where $\check{E}(\bullet|\text{condition}) = E(\bullet|\text{condition}) \cdot P(\text{condition})$ is a shorthand notation used henceforth to denote conditional expectation multiplied with the probability of the condition. The expectations in (30) are taken over the health transitions governed by transition probabilities (7), idiosyncratic component ε_t^p in the out-of-pocket medical shocks (13), and independent idiosyncratic component ε_t^k in wage equation (22). The latter enters the

²⁸We assume that when parameters of the utility function (1) are in the range of interest, the nonnegativity constraint for consumption never binds, i.e. $C_t^p > 0$ unless $w_t^p = 0$.

expression because next period assistance time T_{t+1} depends on the children's circumstances.

The same argument, i.e., the combination of FOCs for f_t and envelope condition, leads to the second Euler equation that characterizes the optimal level of formal insurance f_t , namely

$$0 = \beta^{p} \check{E}_{m,\varepsilon} \left[\frac{\partial u_{t+1}^{p}(m_{t+1}, C_{t+1}^{p}, T_{t+1})}{\partial C_{t+1}^{p}} \left(-(1+r) - \frac{\partial oop_{t+1}(m_{t+1})}{\partial f_{t}} \right) \middle| m_{t+1} \neq \mathbb{D} \right] - \beta^{p} (1+r) p_{m_{t}\mathbb{D}}^{m} (1-\varsigma) u_{b}'(B_{t+1}) - \beta^{p} \sum_{m' \neq \mathbb{D}} \frac{\partial p_{m_{t}m'}^{m}}{\partial s_{t}^{p}} E_{\varepsilon} V_{t+1}^{p}(w_{t+1}^{p}, m', \psi_{t+1}; M_{t+1}, T_{t+1}),$$
(31)

where the expectation is taken over the same random variables. Rearranging and expanding the first expectation using the transition probabilities (7) and plugging in expression (12) for $oop_{t+1}(m_{t+1})$ we get

$$(1+r)\check{E}_{m,\varepsilon}\left[\frac{\partial u_{t+1}^{p}(m_{t+1},C_{t+1}^{p},T_{t+1})}{\partial C_{t+1}^{p}}\middle|m_{t+1}\neq\mathbb{D}\right]+(1+r)p_{m_{t}\mathbb{D}}^{m}(1-\varsigma)u_{b}^{\prime}(B_{t+1})=$$

$$-p_{j\mathbb{D}}^{m}(t)E_{\varepsilon}\left[\frac{\partial u_{t+1}^{p}(\mathbb{D},C_{t+1}^{p},T_{t+1})}{\partial C_{t+1}^{p}}\cdot0\right]$$

$$-p_{j\mathbb{I}}^{m}(t)E_{\varepsilon}\left[\frac{\partial u_{t+1}^{p}(\mathbb{I},C_{t+1}^{p},T_{t+1})}{\partial C_{t+1}^{p}}\cdot\frac{\partial\max\left\{0,\ hc(\mathbb{I})-F_{t+1}(f_{t},\mathbb{I})+\psi_{t+1}\right\}\right]}{\partial f_{t}}\right]$$

$$-p_{j\mathbb{F}}^{m}(t)E_{\varepsilon}\left[\frac{\partial u_{t+1}^{p}(\mathbb{F},C_{t+1}^{p},T_{t+1})}{\partial C_{t+1}^{p}}\cdot\frac{\partial\max\left\{0,\ hc(\mathbb{F})-F_{t+1}(f_{t},\mathbb{F})+\psi_{t+1}\right\}}{\partial f_{t}}\right]$$

$$-p_{j\mathbb{G}}^{m}(t)E_{\varepsilon}\left[\frac{\partial u_{t+1}^{p}(\mathbb{G},C_{t+1}^{p},T_{t+1})}{\partial C_{t+1}^{p}}\cdot0\right]$$

$$-\beta^{p}\sum_{m'\neq\mathbb{D}}\frac{\partial p_{m_{t}m'}^{m}}{\partial s_{t}^{p}}E_{\varepsilon}V_{t+1}^{p}(w_{t+1}^{p},m',\psi_{t+1};M_{t+1},T_{t+1}),$$
(32)

The second and fifth lines in (32) disappear due to the fact that $oop_{t+1}(1) = oop_{t+1}(0) = 0$. Further, it follows from (14) that the partial derivatives in the middle two lines equal either 0 or $-\frac{\omega}{m_{t+1}}$ depending on the sign of the out-of-pocket expenditures. Therefore the expectations in the middle lines can be expressed using conditional expectations as

$$(1+r)\check{E}_{m,\varepsilon}\left[\frac{\partial u_{t+1}^{p}(m_{t+1},C_{t+1}^{p},T_{t+1})}{\partial C_{t+1}^{p}}\middle|m_{t+1}\neq\mathbb{D}\right]+(1+r)p_{m_{t}\mathbb{D}}^{m}(1-\varsigma)u_{b}^{\prime}(B_{t+1})=$$

$$+\frac{\omega}{\mathbb{I}}\check{E}_{\varepsilon}\left[\frac{\partial u_{t+1}^{p}(\mathbb{I},C_{t+1}^{p},T_{t+1})}{\partial C_{t+1}^{p}}\middle|m_{t+1}=\mathbb{I} \text{ and }hc(\mathbb{I})-F_{t+1}(f_{t},\mathbb{I})+\psi_{t+1}>0\right]$$

$$+\frac{\omega}{\mathbb{F}}\check{E}_{\varepsilon}\left[\frac{\partial u_{t+1}^{p}(\mathbb{F},C_{t+1}^{p},T_{t+1})}{\partial C_{t+1}^{p}}\middle|m_{t+1}=\mathbb{F} \text{ and }hc(\mathbb{F})-F_{t+1}(f_{t},\mathbb{F})+\psi_{t+1}>0\right]$$

$$-\beta^{p}\sum_{m^{\prime}\neq\mathbb{D}}\frac{\partial p_{m_{t}m^{\prime}}^{m}}{\partial s_{t}^{p}}E_{\varepsilon}V_{t+1}^{p}(w_{t+1}^{p},m^{\prime},\psi_{t+1};M_{t+1},T_{t+1}),$$

$$(33)$$

and further substituting in the expression for ψ_{t+1} (13) and for $F_{t+1}(f_t, m_{t+1})$ (14) to

$$(1+r)\check{E}_{m,\varepsilon}\left[\frac{\partial u_{t+1}^{p}(m_{t+1},C_{t+1}^{p},T_{t+1})}{\partial C_{t+1}^{p}}\middle|m_{t+1}\neq\mathbb{D}\right]+(1+r)p_{m_{t}\mathbb{D}}^{m}(1-\varsigma)u_{b}^{\prime}(B_{t+1})$$

$$+\beta^{p}\sum_{m^{\prime}\neq\mathbb{D}}\frac{\partial p_{m_{t}m^{\prime}}^{m}}{\partial s_{t}^{p}}E_{\varepsilon}V_{t+1}^{p}(w_{t+1}^{p},m^{\prime},\psi_{t+1};M_{t+1},T_{t+1})=$$

$$=\frac{\omega}{\mathbb{I}}\check{E}_{\varepsilon}\left[\frac{\partial u_{t+1}^{p}(\mathbb{I},C_{t+1}^{p},T_{t+1})}{\partial C_{t+1}^{p}}\middle|m_{t+1}=\mathbb{I} \text{ and }\exp\left(\varepsilon_{t+1}^{p}\right)>\Lambda(\mathbb{I},f_{t})\right]+$$

$$+\frac{\omega}{\mathbb{F}}\check{E}_{\varepsilon}\left[\frac{\partial u_{t+1}^{p}(\mathbb{F},C_{t+1}^{p},T_{t+1})}{\partial C_{t+1}^{p}}\middle|m_{t+1}=\mathbb{F} \text{ and }\exp\left(\varepsilon_{t+1}^{p}\right)>\Lambda(\mathbb{F},f_{t})\right],$$

$$(34)$$

$$\Lambda(m_{t+1}, f_t) = \frac{\overline{F}(m_{t+1}) + \frac{\omega f_t}{m_{t+1}} - hc(m_{t+1})}{\left(\overline{\psi}\right)^{1 - \rho_{\psi}} \psi_t^{\rho_{\psi}}}$$

The resulting expression (34) is an equation with one unknown choice variable in period t, namely f_t , and therefore can be used to compute the optimal level of formal insurance purchase in period t conditional on period t state ($j = m_t, \psi_t$), optimal levels of decision variables in period t + 1 (C_{t+1}^p, T_{t+1}), and state transitions from period t to period t + 1.

Consider next the first corner solution when the credit constraint is the binding constraints, i.e. $w_t^p = C_t^p + f_t$. In this case the problem is much simplified due to the fact that $s_t^p = 0$ and neither the amount of bequest nor health transition probabilities depend on the choices of consumption C_t^p or formal health insurance f_t , and the next period value function depends on these choices only trough the out-of-pocket health spending. Bearing in mind that $\frac{\partial f_t}{\partial C_t^p} = -1$, we have the Euler equation of the form

$$\frac{\partial u_t^p(m_t, C_t^p, T_t)}{\partial C_t^p} = \beta^p \check{E}_{m,\varepsilon} \left[\frac{\partial u_{t+1}^p(m_{t+1}, C_{t+1}^p, T_{t+1})}{\partial C_{t+1}^p} \cdot \frac{\partial oop_{t+1}(m_{t+1})}{\partial f_t} \middle| m_{t+1} \neq \mathbb{D} \right].$$
(35)

Performing similar expansion of the expectation with respect to transition probabilities of health leads to

=

$$\frac{\partial u_t^p(m_t, C_t^p, T_t)}{\partial C_t^p} = \\
= \beta^p \frac{\omega}{\mathbb{I}} \check{E}_{\varepsilon} \left[\frac{\partial u_{t+1}^p(\mathbb{I}, C_{t+1}^p, T_{t+1})}{\partial C_{t+1}^p} \middle| m_{t+1} = \mathbb{I} \text{ and } \varepsilon_{t+1}^p > \Lambda(\mathbb{I}, f_t) \right] + \\
+ \beta^p \frac{\omega}{\mathbb{F}} \check{E}_{\varepsilon} \left[\frac{\partial u_{t+1}^p(\mathbb{F}, C_{t+1}^p, T_{t+1})}{\partial C_{t+1}^p} \middle| m_{t+1} = \mathbb{F} \text{ and } \varepsilon_{t+1}^p > \Lambda(\mathbb{F}, f_t) \right].$$
(36)

Next consider the second corner solution, where the binding constraint is $f_t = 0$. In this case, the Euler equation (30) is derived in the similar way as in the interior solution, and the level of formal insurance is determined by the binding constraint itself.

Finally, in the *third* case, when both the above constraints bind, i.e. $w_t^p = C_t^p + f_t$ and $f_t = 0$, the level of consumption is trivially determined from the constraints. In this case the resources are spent solely for consumption.

To summarize, the interior solution is characterized by the system of equations (30) and (34) which equate marginal utility of current consumption and appropriately scaled expected marginal utility of consumption or bequest in the next period with appropriate adjustments for the effects on health transition probabilities, and further to discounted expected of marginal utility of consumption in the next period conditional on positive out-of-pocket health spending. In the two first corner solutions, either former or latter equality is relaxed, and the solution is characterized by a system of binding constraint and a single equation, namely (30) or (36). In the third case, both of the first order condition equations do not hold, and the solution is uniquely determined by the binding constraints.

A.2 Euler equations in the children's problem

The children's problem is given in the Bellman equation (25). Similarly to the parents problem, first consider *interior solution* case when neither of the constraints is binding.²⁹

 $^{^{29}\}mathrm{We}$ disregard the non-negativity constraint on consumption for the same reasons as in the parents problem, see footnote 28.

Again, standard argument leads to the standard Euler equation

$$\frac{\partial u_t^k(C_t^k, m_t)}{\partial C_t^k} = \beta^k (1+r) E_{m,\varepsilon} \left[\frac{\partial u_{t+1}^k(C_{t+1}^k, m_{t+1})}{\partial C_{t+1}^k} \right],\tag{37}$$

where the expectation is taken over the shocks in the wage equation (22), out-of-pocket health spending (7) and parent's health process including survival.

Following a similar argument, namely the combination of first order conditions and the envelope condition, leads to the Euler equation characterizing the optimal level of transfers. For M_t we have

$$\beta^{k} E_{m,\varepsilon} \left[\frac{\partial u_{t+1}^{k}(C_{t+1}^{k}, m_{t+1})}{\partial C_{t+1}^{k}} \left(-(1+r) + \frac{\partial B_{t+1}(m_{t+1})}{\partial M_{t}} \right) \right] + \beta^{k} \frac{\partial s_{t}^{p}}{\partial M_{t}} \sum_{m'} \frac{\partial p_{m_{t}m'}^{m}}{\partial s_{t}^{p}} E_{\varepsilon} V_{t+1}^{k}(w_{t+1}^{k}, edu; m', s_{t+1}^{p}) = 0.$$

$$(38)$$

The derivative of the bequest (24) can be expressed through the derivative of end-ofperiod assets of the parents s_t^p . Rearranging and expanding the expectation over the health state m_{t+1} we get

$$(1+r)E_{m,\varepsilon}\left[\frac{\partial u_{t+1}^{k}(C_{t+1}^{k},m_{t+1})}{\partial C_{t+1}^{k}}\right] = (1+r)\frac{\partial s_{t}^{p}}{\partial M_{t}}\check{E}_{\varepsilon}\left[\frac{\partial u_{t+1}^{k}(C_{t+1}^{k},m_{t+1})}{\partial C_{t+1}^{k}}\middle| m_{t+1} = \mathbb{D}\right] + \frac{\partial s_{t}^{p}}{\partial M_{t}}\sum_{m'}\frac{\partial p_{m_{t}m'}^{m}}{\partial s_{t}^{p}}E_{\varepsilon}V_{t+1}^{k}(w_{t+1}^{k},edu;m',s_{t+1}^{p}).$$

$$(39)$$

Similarly, for T_t we have

$$E_{m,\varepsilon} \left[W_{t+1} \frac{\partial u_{t+1}^k(C_{t+1}^k, m_{t+1})}{\partial C_{t+1}^k} \right] = (1+r) \frac{\partial s_t^p}{\partial T_t} \check{E}_{\varepsilon} \left[\frac{\partial u_{t+1}^k(C_{t+1}^k, m_{t+1})}{\partial C_{t+1}^k} \middle| m_{t+1} = \mathbb{D} \right] + \frac{\partial s_t^p}{\partial T_t} \sum_{m'} \frac{\partial p_{m_tm'}^m}{\partial s_t^p} E_{\varepsilon} V_{t+1}^k(w_{t+1}^k, edu; m', s_{t+1}^p).$$

$$(40)$$

The resulting expressions (39) and (40) are both equations with one unknown choice variable in period t, namely M_t and T_t , and therefore can be used to compute the optimal level of transfers in period t conditional on period t state, optimal levels of decision variables in period t + 1 (C_{t+1}^p , T_{t+1}), and state transitions from period t to period t + 1. Moreover, the unknowns M_t and T_t only enter in the derivatives of the parent's end of period assets s_t^p , which implies that solving (39) and (40) entails simple search for a point in which the function of parent's end-of-period assets has a particular value of the corresponding partial derivative.

The number of *corner solutions* in the children's problem is higher than in the parent's problem due to the fact that the corner solutions in monetary transfers may coincide with the corner solutions in time transfers. We consider the two sets of corner solutions separately.

The two corner solutions in time transfer are obtained when constraints $T_t = 0$ or $T_t = L$ bind. The level of time transfer is then trivially determined by the binding constraint that replace equation (40). Equations (37) and (39) remain in the system of first order conditions in this case.

The first corner solution in M_t follows from the binding credit constraint $w_t^k = C_t^k + M_t$. Bearing in mind that $\frac{\partial M_t}{\partial C_t^k} = -1$, and again combining first order conditions for C_t^k with envelope condition, we have the following Euler equation for this case

$$\frac{\partial u_t^k(C_t^k, m_t)}{\partial C_t^k} = \beta^k (1+r) \frac{\partial s_t^p}{\partial M_t} \check{E}_{\varepsilon} \left[\frac{\partial u_{t+1}^k(C_{t+1}^k, m_{t+1})}{\partial C_{t+1}^k} \middle| m_{t+1} = \mathbb{D} \right]
+ \beta^k \frac{\partial s_t^p}{\partial M_t} \sum_{m'} \frac{\partial p_{m_tm'}^m}{\partial s_t^p} E_{\varepsilon} V_{t+1}^k(w_{t+1}^k, edu; m', s_{t+1}^p).$$
(41)

The second corner solution in monetary transfer is given by the binding constraint $M_t = 0$ that replaces equation (39). The optimal level of other controls are determined by equations (37) and (40). Finally, in the *third corner solutions* in M_t , both of the above constraints bind leading to $C_t^k = w_t^k$. In this case both equations (37) and (40) are replaced by the binding constraints, so that the level of consumption is equal to the total level of recourses w_t^k , and the optimal level of time transfer is determined by (40) (unless T_t is at the boundary).

To summarize, the otimal decision (C_t^k, M_t, T_t) by children is characterized by the system of three equations, namely (37), (39) and (40) if the solution is in the interior. In addition, there are 6 corner solutions, each characterized by the same system in which one, two or all three equations are replaced by the corresponding binding constraints or equation (41). The structure of the system is similar to that of the parents, however the role of formal health insurance f_t is taken by the monetary transfer M_t , and there is an additional equation to determine the optimal level of assistance time T_t .

A.3 Solution algorithm

Despite that the system of 5 first order conditions (28) is rather complex, and is further complicated by the numerous potential corner solutions, we are able to solve it relatively quickly by implementing the endogenous grid point approach in the following computational algorithm:

- 1. Compute the policy function in the terminal game when t = 100. This is the base for backward induction.
- 2. Begin/continue the outer loop over period $t = t + \Delta$ discrete and discretized state points $(m_t, \psi_t, edu, \Delta)$. For each point:
- 3. Compute solution for the parents problem conditional on a grid of possible values of T_t and s_t^k (parent's best response). Namely:
 - (a) Fix an auxiliary grids over potential current period assistance time T_t and over children's end-of-period wealth s_t^k . For each point:
 - (b) Fix a grid over parent's end-of-period wealth s_t^p . For each point:
 - (c) Numerically solve the system of two Euler equations for the parent's problem (28.a, 28.b) taking into account 3 potential corner solutions as defined by the first column in (29).

The system is solved in the following way. First we solve equation (28.b) which has one unknown current decision f_t (because current consumption does not enter the next period budget other than through s_t^p which is fixed at previous step). Given the end-of-period wealth s_t^p and s_t^k , which enter into the intertemporal budget constraints (16) and (23), the next period state Ω_{t+1} is known up to the random transitions of m_t and idiosyncratic shocks ε_t^p and ε_t^k . We discretize the distribution of the shocks using Gaussian quadrature³⁰ and use transition probabilities (7) of m_t to compute the expectations that enter the equation. For each plausible point in Ω_{t+1} the choice variables, namely C_{t+1}^p , T_{t+1} and M_{t+1} are given by the policy functions computed on the previous

³⁰We make three sets of quadrature points for ε_t^p in order to preserve the accuracy of integration in the conditional expectations.

iteration of the backward induction loop started in step 2. Thus, equation (28.b) has only one unknown and can be solved independently.

We then solve equation (28.a) using exact same procedure, given the found optimal level of formal insurance f_t that enters the intertemporal budget (16). One unfortunate deviation from the Carroll (2004) is that the marginal utility function is not invertible in C_t^p , so the equation also has to be solved numerically.

- (d) One corner solution case requires special treatment. When the credit constraint binds $(s_t^p = 0)$ but the level of formal insurance is strictly positive, i.e. first case in first column in (29), in step 3b the grid is fixed over f_t instead of s_t^p , and the next step is performed the same way.
- (e) Once the optimal levels of current purchase of formal insurance f_t and current consumption C_t^p are found, the amount of consumable resources to which they are attributed is computed using intra-temporal budget (15) completely analogously to the original EGM method. Thus, $w_t^p + M_t = s_t^p + C_t^p + f_t$, and the triplet $(w_t^p + M_t, s_t^p, C_t^p)$ is recorded in the memory as one of potential optimal points conditional on the grid values of step 3a and the amount of monetary transfers M_t .
- (f) For the low values of parents wealth w_t^p and children's transfers M_t it is not clear in which order the three corner solutions follow. Presumably, for very low wealth it holds $C_t^p = w_t^p$, $s_t^p = f_t = 0$; for large values of wealth, the interior solution is most likely. In any case, to distinguish between the relevant corner solutions, we invoke the upper envelope calculation procedure which compares value functions implied by the different solutions, and computes the the thresholds separating them.
- (g) Proceed with the next point in steps 3b and 3a until the points are exhausted.
- 4. With the parent's solution at hand, and in particular the optimal reaction function $s_t^p (w_t^p + M_t, T_t, s_t^k)$, which is calculated on the fixed grid over (T_t, s_t^k) and the endogenous grid over $(w_t^p + M_t)$, proceed to the solution of children's problem. Namely,

- (a) For every point $(w_t^p + M_t)$ of the endogenous grid produced in step 3:
- (b) Fix a grid over children's end-of-period wealth s_t^k , preferably the same grid as in step 3a. For each point:
- (c) Numerically solve the system of three Euler equations for the children's problem (28.c, 28.d, 28.e) taking into account 11 potential corner solutions as defined by the second and third columns in (29).

The system is solved in a similar way to the one in parent's problem. First we solve equation (28.e) which has one unknown current decision T_t (because current consumption and monetary transfers do not enter the next period budget other than through s_t^k which is fixed at previous step). A given trial value T_t together with $(w_t^p + M_t)$ and s_t^k determine parent's purchase of formal insurance f_t and end-of-period wealth s_t^p . On the other hand, the trial value T_t affects the next period wealth of the children. Thus, the next period state Ω_{t+1} is known up to the random transitions of m_t and shocks ($\varepsilon_t^p, \varepsilon_t^k$). We again use two dimensional Gaussian quadrature for the shocks and transition probabilities (7) of m_t to compute the expectations that enter the equation. For each plausible point in Ω_{t+1} the choice variables, namely C_{t+1}^k which enters marginal utility, is given by the policy functions computed on the previous iteration of the backward induction loop started in step 2. Thus, equation (28.e) has only one unknown and is solved independently.

Equations (28.c) and (28.d) can then be solved in any order. Given the optimal choice of assistance time T_t , the right hand side in (28.c) can be computed directly, and because the marginal utility of children's consumption is invertible, the optimal C_t^k is computed directly.

Finally, equations (28.d) entails finding the point M_t at which the derivative $\frac{\partial s_t^p}{\partial M_t} = \frac{\partial s_t^p}{\partial (w_t^p + M_t)}$ takes a particular value, which can be calculated directly. The derivative is approximated from the solution for the parent's problem, given already known T_t and fixed s_t^k . Because the computed policy functions for the parent's problem are not conditional on M_t , the solution is determined up to a constant, namely only $(w_t^p + M_t)$ is determined.

- (d) Similar to the parent's problem, one corner solution case also requires special treatment here. When credit constraint binds $(s_t^k = 0)$ but the level of mone-tary transfer is strictly positive, i.e. first case in third column in (29), in step 4b the grid is fixed over M_t instead of s_t^k , and the next step is performed the same way.
- (e) Once the optimal levels of the "aim" for consumable resources for parents $(w_t^p + M_t)$ are computed, corresponding level of children's resources is computed using intra-temporal budget $w_t^k = s_t^k + C_t^k + M_t$. Even though M_t is not determined exactly, the vector $(w_t^p + M_t, w_t^k + w_t^p, C_t^k, M_t^k, T_t^k)$ is recorded in the memory as one of potential optimal points conditional on the grid values of step 4a and the amount of monetary transfer M_t .
- (f) Again, to distinguish between the relevant corner solutions, we invoke the upper envelope calculation procedure which compares value functions implied by the different solutions, and computes the thresholds separating them.
- (g) Proceed with the next point in steps 4b and 4a until the points are exhausted.
- 5. Combine the solutions of parent's and children's problems through and additional grid on $M_t > 0$. For each point M' on this grid, compute the corresponding values of parent's and children's wealth from the endogenous grid points $(w_t^p + M_t, w_t^k + w_t^p)$ found in steps 3 and 4, i.e. $w_t^p = w_t^p + M_t M', w_t^k = (w_t^k + w_t^p) (w_t^p + M_t M'),$ and associate them with the corresponding values of the choice variables from the solutions found on these grids. This concludes the computations for given point in the discretized state space.
- 6. Proceeds with the next iteration of the backward induction started in step 2 until t = 65 is reached.

B Health and Funeral Costs

When no country data on long term care or curative and rehabilitation care was available, we used the benchmark countries below. The primary data for funeral costs in the OECD countries analyzed are drawn from the AGIR dataset (Westerhout and Pellikaan 2005, based on EPC 2001) for EU-15 countries and from OECD calculations for 2005. To obtain the 2004 funeral costs, we applied the health expenditure real growth rate to the 2005 series (OECD Health Data 2008). The cost of death for the oldest group (95+) is assumed to be the lowest and was proxied by their observed health expenditure per person, when available. For France, Germany, Italy, Spain and Netherlands, where expenditure data for the oldest group were not available, the cost of people aged 75-79 was taken as a proxy. In fact, when available, expenditure at age 95+ is roughly equal to the level of expenditure at age 75-79. For the countries with no data available, the cost of death for the oldest group was estimated by taking three times the average health expenditure per capita, adjusted by the country-specific residual (Bjornerud et al., 2005; OECD, 2006).

Country estimated	Benchmark countries
Belgium	Netherlands
Denmark	average (Norway, Sweden)
France	Germany
Greece	Spain
Italy	average (Germany, Spain)
Switzerland	Germany

		Formal Sector								Informal Sector#			
	Average Health Care Costs per Capita (€)	Public Sector		Private Sector					Money		Time		
Country		Public Expendi- ture+	% of Populatio n with Public	Private Medical Insurance +	Average Premium per Insured (€)	Benefits paid+	Claims Ratio§	% of Population with Private Insurance	Out-of- Pocket Costs +^	% Receiving Transfers from Family	Median Amount from Family (€)	% Receiving Transfers from Family	Median Amount from Family (hours)
Austria	2,658	73.8%	98.0%	3.2%	529	4.5%	86%	31.1%	16.9%	6.0%	800.0	32.2%	580
Belgium	2,639	75.6%	99.0%	4.9%	119	1.7%	94%	44.1%	17.6%	1.5%	1250.0	41.1%	308
Switzerland	4,527	58.5%	100.0%	8.7%	2,370	8.7%	76%	22.9%	31.9%	5.9%	847.4	23.7%	372
Germany	2,723	73.4%	89.8%	9.3%	1,087	9.3%	124%	29.6%	19.2%	3.8%	600.0	36.6%	886
Denmark	3,166	84.1%	100.0%	1.5%	n.a.	6.6%	n.a.	n.a.	14.0%	2.2%	1344.3	31.4%	116
Spain	1,565	70.1%	99.5%	5.9%	373	4.5%	76%	25.6%	23.0%	5.6%	1532.0	22.1%	1,716
France	2,317	78.5%	99.9%	13.2%	451	3.1%	79%	21.9%	21.2%	2.1%	1500.0	34.9%	846
Italy	1,921	75.0%	100.0%	*(2006) 0.9%	n.a.	1.0%	80%	*(2006) 6.1%	21.3%	0.9%	578.4	23.4%	782
Netherlands	2,321	70.1%	71.2%	13.2%	590	18.1%	83.3%	87.3%	18.8%	2.4%	2250.0	32.6%	374
Greece	^1,608	^61.8%	100.0%	*(2002) 1.6%	n.a.	n.a.	n.a.	*(2002) 12%	5.9%	16.8%	821.3	44.0%	1,250
Sweden	2,837	82.3%	100.0%	*(2006) 0.3%	417	0.1%	n.a.	1.9%	17.0%	3.3%	762.5	38.0%	87

Table 1. Health Care in Europe, Sources of Funding

+ As percent of total health care expenditure. § As percent of premium.

Sources: CEA Statistics No. 41: The European Health Insurance Market in 2008, ^ OECD Health Data: Health expenditure and financing,

* Thomson and Mossialos (2009), # Share 2004 data (singles, age 65+, transfers received annually)

	Prevalenc	e (%) of	Premium pe	r insured (€)	Individual OOP costs (€)		
Country	Formal premium*	OOP costs^	Median	Mean	Median	Mean	
Sweden	1.8	91.7	960.00	1961.74	2600.00	4436.66	
Denmark	26.6	86.1	2000.00	1995.74	2000.00	3497.34	
Germany	17.5	88.0	800.00	1869.34	140.00	357.01	
Netherlands	74.3	46.8	440.00	741.82	226.10	545.00	
Belgium	70.9	96.1	181.70	260.58	400.00	893.29	
France	85.2	37.3	715.80	804.05	120.00	409.53	
Switzerland	49.7	74.5	450.00	1296.52	460.00	1328.40	
Austria	21.3	73.0	600.00	912.50	175.00	459.29	
Italy	3.3	77.1	400.00	406.13	200.00	429.82	
Spain	6.4	28.1	700.00	831.70	138.70	485.07	
Greece	3.2	90.7	309.40	533.27	256.50	624.14	
Mean across Europe	32.7	71.8	686.99	1055.76	610.57	1224.14	

Table 2. Formal Premium and Out-of-Pocket Medical Spending - SHARE

* Share of people insured privately in the full sample

^ Share of people with positive out-of-pocket medical spending in the full sample

	Prevalence	e (%) of FT	Amount o	f€ received	Amount of € from kids		
Country	from kids*	from kids in total FT^	Median	Mean	Median	Mean	
Sweden	2.3	60.0	5,000.00	26,562.13	7,000.00	9,548.53	
Denmark	1.5	57.1	9,000.00	13,494.64	9,000.00	10,670.70	
Germany	1.9	54.5	600.00	987.19	500.00	1,239.74	
Netherlands	1.8	71.4	894.70	1,473.98	894.70	973.79	
Belgium	0.6	37.5	1,671.60	7,168.27	500.00	1,072.10	
France	1.0	35.7	2,016.00	2,589.76	2,016.00	2,683.31	
Switzerland	2.7	57.1	1,126.00	2,078.76	1,300.00	2,829.27	
Austria	5.1	69.0	500.00	2,128.61	480.00	911.60	
Italy	1.6	66.7	350.00	437.09	532.90	584.90	
Spain	4.9	81.3	1,087.80	1,808.47	1,000.00	1,737.12	
Greece	15.1	87.5	950.00	1,264.12	950.00	1,302.43	
Mean across Europe	3.5	61.6	2,108.74	5,453.91	2,197.60	3,050.32	

 Table 3. Financial Transfers (FT) - SHARE

* Share of people receiving positive financial transfers from kids in the full sample

^ Share of people receiving positive financial transfers from kids in the sample of people receiving financial transfers

	Prevalence	e (%) of TT	Amount of h	ours received	Amount of hours from kids		
Country	from kids*	from kids in total TT^	Median	Mean	Median	Mean	
Sweden	27.0	61.5	53	261	80	284	
Denmark	25.5	57.4	60	344	60	157	
Germany	30.8	62.5	220	669	223	754	
Netherlands	20.0	46.3	52	202	52	178	
Belgium	28.1	55.1	116	424	104	359	
France	23.8	59.5	156	551	104	457	
Switzerland	17.4	52.0	52	126	50	105	
Austria	21.9	57.0	260	608	260	435	
Italy	13.9	47.9	312	917	377	1240	
Spain	17.2	56.8	260	732	262	711	
Greece	39.3	72.8	275	709	208	517	
Mean across Europe	24.1	57.2	165	504	162	472	

Table 4. Time Transfers (TT) - SHARE

* Share of people receiving positive time transfers from kids in the full sample

^ Share of people receiving positive time transfers from kids in the sample of people receiving time transf